Author's response to reviews

Title: Chronic pain treatment in children and adolescents: Less is good, more is sometimes better

Authors:

Tanja Hechler (t.hechler@deutsches-kindenschmerzzentrum.de)
Julia Wager (j.wager@deutsches-kindenschmerzzentrum.de)
Boris Zernikow (b.zernikow@kinderklinik-datteln.de)

Version: 3
Date: 30 September 2014

Author's response to reviews: see over
Re: Chronic pain treatment in children and adolescents: Less is good, more is sometimes better
Tanja Hechler¹, PhD, Julia Wager¹, PhD, Boris Zernikow¹, MD, PhD

Dear Professor O'Donovan,

We are very happy that our manuscript is now been considered suitable for publication in *BMC Pediatrics*.

As requested, we have included the email addresses of all authors on the title page. In addition, we have included the Tables in the manuscript in a Tables section following the references.

Yours sincerely (on behalf of the authors)

Tanja Hechler

Head of the Research Team
German Paediatric Pain Centre and
Department of Children’s Pain Therapy and Paediatric Palliative Care,
Children’s and Adolescents’ Hospital Datteln,
Witten/Herdecke University